Author's response to reviews

Title: Monophasic Synovial Sarcoma presenting as a primary ileal mass: Case Report and Literature Review.

Authors:

Alaa N Alsharif (ahmadtafigue@yahoo.com)
Musa M Fageeh (yousofaaa@aol.com)
Yousof A Alabdulkarim (Yousof.abdulkarim@gmail.com)

Version: 7 Date: 29 October 2011

Author's response to reviews:

- Please include a title page as page 1. The title page should include the title, authors, author affiliations and corresponding author details.
  A- a title page was included containing required details

- In keeping with the journal’s style, please remove subheadings under the Case Presentation Section.
  A- subheadings were removed

- Please replace the header “Case Report” to “Case Presentation.”
  A- the sentence was corrected

thank you